
SAFETY NOTES

Please pay attention to additional safety notes as described in detail in the User Manual, Chapter 2, 
„SAFETY NOTES AND LABELS“.

THE INSTRUMENT

8“ color Touch screen

Status indicator LED's

USB ports

Sample inlet

Air exhaust

Fan cover

CF card cover

USB, COM ports

Network connection

front view back view

ON/OFF switch
(rear panel andbehindfront door)

INSTALLATION

Installation example:
Aethalometer AE33, Sample Stream Dryer, cyclone and MET station 
Sample Stream Dryer, Cyclone, MET station - OPTIONAL

Any water intake may damage  the instrument. 
Damage due to water intake is not covered by the warranty! 

1) Remove red protective caps from the

instrument.

2) Attach the sample inlet line using

fittings and adapters. Avoid sharp bends.

3) Switch the instrument on (make sure 

to switch on TWO switches, one on rear panel, 

one behind front door).

4) Protect the inlet from rain using a cyclone. 

Ensure periodical cleaning!

5) Install Sample Stream Dryer especially in cases when outside 

Dew point is higher than the temperature in the station.

6) Avoid direct discharge of air-conditioner over 

sample lines, or insulate them.  (This may cause 

water condensation in the tubing).

7) Install insect screen/water trap (provided

with the instrument). The cup must be 

oriented downwards.

The instrument must be protected from outdoor weather, and operated with ambient temperature 
from 10 to 40°C. Avoid direct effect of air-conditioners on the instrument or sampling lines.
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SAMPLE STREAM DRYER

CYCLONE AMBIENT METEOROLOGICAL SENSOR

VACUUM PUMP

AE33

Avoid direct discharge of air-

conditioner over sample lines.



M
a

g
e

e
 S

ci
e

n
tif

ic
 A

e
th

a
lo

m
e
te

r 
A

E
3
3

 
Q

U
IC

K
 S

T
A

R
T

 GU
ID

E

Date/Time settings
Basic measurement settings
Test procedures buttons

Internal status codes/values

MAC (m2/g) for all 7 channels

Ångstrom exponents (source apportionment)

Current SW/FW version
Compensation algorithm parameters

Calibration/Adjustment buttons
Procedures

Display settings

Compensation parameters
Black Carbon concentration (BC)
Compensated values of BC concentration
(370, 470, 520, 590, 660, 880 and 950 nm)

Flow through Spot1/Spot2

Black Carbon concentration (BC6)

UVPM (BC1) or BC biomass burning [%]

Measured flow rate (2, 3, 4 or 5 LPM)

Time base (1 or 60 seconds) 

Estimated number of tape advances until the
end of tape

Status of the instrument (press the icon 
for more information)

HOME SCREEN AND USER INTERFACE DESCRIPTION

BASIC STATUS CODES

STATUS NO DESCRIPTION 4 (8, 12) Flow issues

0 Measurement running 128 (256, 384) Tape warning

1 (2) Tape advance (first measurement) 1024 (2048, 3072, 4096, 
6144 

Various tests (service and maintenance procedures)

3 Measurement stopped 8192 Connection error (external device)

START/STOP measurement



MAINTENANCE PROCEDURE INTERVAL

Procedure(refer toUser Manual for details) 1 month 6 months 12 months

Check sample inlet flow

Inspectsample line tubing

Inspect/clean size selective inlet/cyclone/insect screen

Verifytime and date

Inspect/clean optical chamber

Validate/calibrate flow, perform leakage test

Perform Clean Air/Stability checktest

Perform ND test

Lubricateoptical chamber sliders

Calibratefilter tape optical sensors

Install new filter tape roll As needed

Change cartridgefilter As needed

Filter tape exchange

1. STOP the measurements, open the front door. 

2. Loosen the thumb screws and remove the transparent plastic covers from both the left-hand and the right-hand side 
spool.
3. Using thumb and fingers on both sides of the optical chamber, lift the chamber upwards against its springs and lock it in 
place by pushing the metal latch. Remove the used tape from both axles. 

5. Take the new filter tape and place it on the left axis. Take note of the metal spring leaf which stabilizes the top surface of 
the roll. Take the other cardboard spool and attach the end of the filter to it using adhesive tape. Place the empty on the 
right spool axis. 

6. Make sure that the tape is fed beneath both the left and the right guide posts, and passes under the optical chamber. 
Release the optical chamber by lifting up with thumbs and fingers on both sides. 

7. Replace both transparent plastic covers. Tighten the thumb screws by hand only. Check that the left spool can turn 
freely. Make sure that the right spool is firmly clamped to its axis: test by gently pulling the tape, and make sure the spool 
center does not move. 

*Go to „OPERATION-ADVANCED“ screen, press „Change Tape“ button and follow the instructions on the screen 
or follow step-by-step procedure described below.
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REGISTER YOUR PRODUCT
In order to receive latest updates, notifications about new features and functionalities, access to 
all technical materials (spec sheets, technical papers, video tutorials) and other resources, we 
encourage all users to complete the registration form at http://www.mageesci.com/register

The Users Group site offers interactive resources at www.mageesci.com/group. 

CONTACT INFORMATION

AEROSOL (globally)
T: +386-1-4391-700
support@aerosol.eu 

Magee Scientific (USA,MEX,CAN)
T: +1-510-845-7137
support@mageesci.com

Cleaning the optical chamber

1. STOP the instrument, SHUT DOWN and switch off the power.

2. Using thumb and fingers on both sides of the optical chamber, lift the chamber upwards against its springs. When the 
chamber is fully raised, lock it in place by pushing in the metal latch. 

3. Find the release button on the front of the chamber. While pressing the release button upward, grasp the lower 
portion of the optical head and turn it clockwise so the front of the chamber moves to the left. 

4. The lower portion of the optical head consists of a translucent block. Clean all surfaces with ethanol. Put a few drops 
of ethanol in the openings and remove any dust or debris by using soft cotton swabs. 

5. Use compressed air to remove any debris (fibres) from the openings. No material shouldbe left on/in the openings.

6. To re-install the optical head, first align the notched marker to the left of center; push the optical head upwards to 
engage the three bayonet fittings; and turn counter-clockwise (so the front of the chamber moves to the right). 

7. Using thumb and fingers on both sides of the optical chamber, lift the chamber upwards. The locking latch will 
automatically release. Allow the chamber to return down to the tape. 

More details in the User Manual, Chapter 9,- „MAINTENANCE“.

http://www.mageesci.com/register
http://www.mageesci.com/group

